Author's response to reviews

Title: A cluster randomised controlled trial evaluating the effectiveness of a structured pulmonary rehabilitation education programme for improving the health status of people with chronic obstructive pulmonary disease (COPD): The PRINCE Study.

Authors:

Kathy Murphy (kathy.murphy@nuigalway.ie)
Dympna Casey (dympna.casey@nuigalway.ie)
Declan Devane (declan.devane@nuigalway.ie)
Adeline Cooney (adeline.cooney@nuigalway.ie)
Bernard McCarthy (bernard.mccarthy@nuigalway.ie)
Lorraine Mee (lorraine.mee@nuigalway.ie)
Martina Nichulain (martinanichulain@nuigalway.ie)
Andrew W Murphy (andrew.murphy@nuigalway.ie)
John Newell (john.newell@nuigalway.ie)
Eamon O’ Shea (eamon.oshea@nuigalway.ie)

Version: 5 Date: 17 January 2011

Author's response to reviews:

final proof now completed